

Response to Reviewers #1

Manuscript: “Working Title of the Paper”

First Author^a, Second Author^b, and Third Author^a

^a*First Institute, First University, Country*

^b*Second Institute, Second University, Country*

This response to the review comments adheres to the following style code:

- *Reviewer comments are quoted verbatim, in italics.*
- Our answers follow in upright text.
- Revised manuscript text is quoted in blue.

Line numbers refer to the annotated version of the revised manuscript.

Response to Editor

Comment E.1

Quote the editor's request here.

Answer it, and say where in the manuscript the change landed (line 2).

Reviewer 1

Comment 1.1

Quote the first comment verbatim, so the reviewer can recognise it.

Answer the comment directly. If the manuscript changed, point at the line and quote the new text.

The revised sentence, copied from the manuscript (line 14).

Comment 1.2

Quote the second comment.

Where a comment is declined, say so plainly and give the reason rather than deferring it.

Reviewer 2

Comment 2.1

Quote the comment.

Answer it (line 25).

Comment 2.2

Quote the comment.

Answer it (line 34).